Author’s response to reviews

Title: Gender differences in acute recreational drug toxicity: a case series from Oslo, Norway

Authors:

Victoria Syse (victoriasyse@yahoo.no)
Mette Brekke (mette.brekke@medisin.uio.no)
Marit Grimsrud (m.m.grimsrud@medisin.uio.no)
Per Sverre Persett (p.s.persett@studmed.uio.no)
Fridtjof Heyerdahl (frihey@gmail.com)
Knut Erik Hovda (knuterikhovda@gmail.com)
Odd Martin Vallersnes (o.m.vallersnes@medisin.uio.no)

Version: 3 Date: 08 Apr 2019

Author’s response to reviews:

Dear editors

Thank you for this opportunity to revise our manuscript.

- Please rename Objectives to Background in the Abstract section.

We have renamed this subsection as requested.

- Please amend your 'Ethics approval and consent to participate' statement as currently there are two conflicting statements- 'the need for approval from an ethics committee was waived' and 'The study was approved by the Oslo University Hospital Information Security and Privacy Office (ref no 2013/3706), and by the director of the Department of Emergency General Practice at the City of Oslo Health Agency.'

We have amended the Ethics approval and consent to participate statement, as requested.
Oslo, 8 April 2019

On behalf of the authors

Yours sincerely

Odd Martin Vallersnes

Associate professor
Department of General Practice
University of Oslo

Consultant General Practitioner
Oslo Accident and Emergency Outpatient Clinic
Department of Emergency General Practice
City of Oslo Health Agency